©American Psychological Association, 2017. This paper is not the copy of record and may not exactly replicate the authoritative document published in the APA journal. Please do not copy or cite without author's permission. The final article is available, upon publication, at: doi: 10.1037/ccp0000220. This article may not exactly replicate the final version published in the APA journal. It is not the copy of record.

Do Changes in Trauma-related Beliefs Predict PTSD Symptom Improvement in Prolonged Exposure and Sertraline?

Accepted Version as Submitted April 4, 2017

Andrew A. Cooper ^{a*}, Lori A. Zoellner ^b, Peter Roy-Byrne ^c,

Matig R. Mavissakalian ^d, & Norah C. Feeny ^a

^a Department of Psychological Sciences, Case Western Reserve University, Cleveland, OH, USA

^b Department of Psychology, University of Washington, Seattle, WA, USA

^c Department of Psychiatry, University of Washington, Seattle, WA, USA

^d Department of Psychiatry, Case Western Reserve University, Cleveland, OH, USA

^{*}Dr. Cooper is now with the Department of Psychology, University of Toronto Scarborough. Please address all correspondence to Andrew A. Cooper, University of Toronto Scarborough, Department of Psychology, 1265 Military Trail, Scarborough, Ontario, Canada, M1C 1A4. Email: aa.cooper@utoronto.ca

Preparation of this manuscript was supported by grants to Drs. Zoellner and Feeny from the National Institute of Mental Health (R01 MH066347, R01 MH066348) and the William T. Dahms, MD, Clinical Research Unit, funded under the Cleveland Clinical and Translational Science Award (UL1 RR024989). Sertraline was provided free of charge by Pfizer, Inc.

Public Health Significance Statements

Changing trauma-related negative beliefs may be an important pathway to PTSD symptom improvement for cognitive behavioral psychotherapies such as prolonged exposure (PE) but is perhaps less central for pharmacotherapy using selective serotonin reuptake inhibitors (SSRIs). Consistent with specificity to PTSD, there was less evidence for the role of change in trauma-related beliefs for depression symptoms. Processes that promote symptom improvement may differ between PE and SSRIs whereby change in PE may occur through specific learning processes surrounding adaptively shifting views about oneself, the world, and others.

Abstract

Objective: Negative trauma-related belief change has been found to predict subsequent improvement in symptoms of posttraumatic stress disorder (PTSD) in prolonged exposure (PE) and other therapies, consistent with several psychological theories of treatment change (e.g., Foa & Kozak, 1986). However, belief change has not been examined in selective serotonin reuptake inhibitors such as sertraline. We examined processes associated with symptom improvement in two treatments for PTSD, hypothesizing that belief change would robustly predict PTSD improvement in patients treated with PE but not those treated with sertraline, reflecting moderation by treatment. **Method:** Patients with chronic PTSD (N = 134; 78% female, 71.6% Caucasian, M = 38.1 years, SD = 11.8) received 10 weeks of PE or sertraline in a randomized controlled trial. Patients reported PTSD and depression symptoms, and trauma-related beliefs (Post-Traumatic Cognitions Inventory; Foa et al., 1999) at pre-treatment, every treatment session, and post-treatment. **Results:** Using time-lagged mixed regression models, change in trauma-related beliefs predicted subsequent PTSD symptom improvement, an effect moderated by treatment and particularly strong in PE (d = 0.93) compared to sertraline (d = 0.35). Belief change also predicted depressive symptom improvement but more modestly and bi-directionally, with no difference by treatment modality. *Conclusions:* Trauma-related belief change precedes PTSD improvement more robustly in PE than sertraline and with greater specificity compared to

depressive symptoms. These findings highlight potentially divergent processes contributing to symptom change in these PTSD treatments, with belief change as a key mechanism of PE. *Keywords:* PTSD, exposure, selective serotonin reuptake inhibitors, sertraline, beliefs, process of change

Do Changes in Trauma-Related Beliefs Predict PTSD Symptom Improvement in Prolonged

Exposure and Sertraline?

The past three decades of research on the treatment of mental health disorders has yielded a variety of empirically-supported psychotherapeutic and pharmacological treatments (Nathan & Gorman, 2015). However, we still know relatively little about how these interventions work, in terms of the mechanisms of acute and enduring symptom improvement (Kazdin, 2007; Murphy, Cooper, Hollon, & Fairburn, 2009). As stated succinctly by McNally (2007): "Theoretical agnosticism about mediating mechanisms is acceptable only when treatment works with flawless fidelity" (p. 751). Persistent evidence across a variety of disorders of incomplete or non-response to treatment (e.g., Hirschfeld et al., 2002; Taylor, Abramowitz, & McKay, 2012), as well as relapse and recurrence after treatment (e.g., Bruce et al., 2005; Mueller et al., 1999), underscores the need to move beyond theory and toward empirical tests of potential treatment mechanisms. By identifying mechanisms linked to symptom improvement, researchers and clinicians can maximize the likelihood of successful and effective treatment across patients (Murphy et al., 2009). Mechanism research has been highlighted as a research funding priority (Cuthbert & Insel, 2013), with a particular emphasis on integration of psychological and neurobiological indices and identification of common processes across related disorders and treatments (e.g.,

Hofmann, Ellard, & Siegle, 2012).

Cognitive behavioral therapies (CBTs) are empirically-supported for many disorders (Hofmann, Asnaani, Vonk, Sawyer, & Fang, 2012) and dominate the literature on theoretical mechanisms of psychotherapeutic symptom change. Although specific treatment protocols vary both across and within disorder, CBT for anxiety and traumatic stressor disorders commonly involve some form of structured exposure exercises. Contemporary theories of exposure-based treatments draw on the behavioral and learning roots of these therapies in emphasizing altering the meaning of conditioned associations (conditioned stimuli [CS]-unconditioned stimuli [US]) as a central mechanism of treatment, fostering new inhibitory learning (e.g., Bouton, 1991, 2004; Craske, Treanor, Conway, Zbozinek, & Vervliet, 2014; Foa, Huppert, & Cahill, 2006). Consistent with these models, many contemporary theories posit that changes in disorder-specific negative or distorted beliefs may act as potential mechanisms of the observed treatment effects (e.g., Beck, Emery, & Greenberg, 2005; Hoffman, 2008).

In the case of posttraumatic stress disorder (PTSD), negative trauma-related beliefs about oneself, others, and the world have a prominent role in many psychological models (Ehlers & Clark, 2000; Foa, Huppert, & Cahill, 2006; for a recent review, see LoSavio, Dillon & Resick, 2017). Some studies have shown that peritraumatic endorsement of these beliefs predicts subsequent development of PTSD (e.g., Ehring, Ehlers, & Glucksman, 2008; Shahar, Noyman,

Schindel-Allon, & Gilboa-Schechtman, 2013), as well as severity and maintenance of PTSD symptoms (e.g., Bryant & Guthrie, 2007; Dunmore, Clark, & Ehlers, 1999; Foa & Cahill, 2001; Lancaster, Rodriguez & Weston, 2011). The interplay between beliefs and PTSD symptoms has been characterized as a "vicious cycle" leading to worsening impairment over time (Shahar et al., 2013). Reductions in these beliefs occur across successful PTSD treatments with a variety of cognitive-behavioral psychotherapies (Foa & Rauch, 2004; Kleim et al., 2013; Resick et al., 2002; Taylor et al., 2003; Zoellner, Feeny, Efthekari, & Foa, 2011). Changes in trauma-related beliefs have therefore been emphasized as an important target of some psychological treatments for PTSD as a potential mediator of symptom change (LoSavio et al., 2017).

Prolonged exposure therapy (PE) is a gold standard treatment for PTSD with strong empirical support across a variety of samples and trauma types (Powers, Halpern, Ferenschak, Gillihan, & Foa, 2010). PE was developed based on earlier behavioral therapies and on the foundation of emotional processing theory (Foa & Kozak, 1986). Historically, emotional processing theory has primarily emphasized behavioral and learning models of fear conditioning and extinction (e.g., Lang, 1979). However, recent updates have more clearly highlighted the importance of change in trauma-related beliefs as a key process in recovery from PTSD (Foa et al., 2006; Foa & McLean, 2016). In PE, change in negative thinking patterns is thought to be facilitated by providing disconfirming information through exposure exercises. For instance, a

belief that one is weak and incompetent might be counteracted by evidence of mental strength and skillfulness, as demonstrated through repeatedly recounting one's life-saving actions throughout the trauma. Several studies have shown that negative trauma-related cognitions are substantially reduced after PE and correlated with PTSD symptom change (e.g., Foa & Rauch, 2004; Hagenaars, van Minnen, & de Rooij, 2010; Nacasch et al., 2015). Trauma-related belief change has also been found to mediate symptom change in some (Mueser et al., 2008; Nacasch et al., 2015; Zoellner et al., 2011) but not all (Hagenaars et al., 2010) studies of PE. Careful assessment of the temporal sequence of cognitive and symptom change is critical to ascertaining whether belief change might "drive" symptom improvements (e.g., Kazdin, 2007; Murphy et al., 2009) in PE and other PTSD interventions. Using time-lagged, repeated measures analyses, a series of recent studies have found that belief change predicts subsequent improvements in PTSD symptoms in CBT treatments for adults (Kleim et al., 2013; Kumpula et al., 2016; McLean, Su, & Foa, 2015; Oktedalen, Hoffart & Langkaas, 2015; Schumm, Dickstein, Walter, Owens & Chard, 2015; Zalta et al., 2014) and adolescents (McLean, Yeh, Rosenfield, & Foa, 2015). Most of these studies have shown that belief change precedes symptom improvement, with just one showing a more reciprocal relationship between beliefs and symptoms (McLean, Su et al., 2015). Thus, there is preliminary evidence that trauma-related belief change precedes, and perhaps drives, symptom improvement in CBT for PTSD. This is consistent with the view of belief

change as a process which underlies successful recovery from PTSD, regardless of whether this is achieved via PE, another treatment, or naturalistic improvement (Foa et al., 2006). Consistent with this theory, belief change has been found to precede PTSD improvement in other empirically-supported psychotherapies, including other CBTs like Cognitive Processing Therapy (Kleim et al., 2013; Schumm et al., 2015) and non-CBT approaches that are not clinically or theoretically focused on modifying trauma-related cognitions (e.g., McLean, Yeh et al., 2015; McLean, Su et al., 2015).

Selective serotonin reuptake inhibitors (SSRIs) are efficacious in the treatment of a variety of mental health disorders, including unipolar mood disorders (e.g., major depressive disorder [MDD]; Hollon et al., 2005) and PTSD (Jonas et al., 2013; Lee, Schnitzlein, Wolf, Vythilingam, Rasmusson, & Hoge, 2016). One contemporary model of the therapeutic action of SSRIs posits that they intiate a cascade of intracellular effects that ultimately ameliorate a broad range of neuronal deficits most typified by decreased neurogenesis and disrupted neuroplasticity (Duman & Voleti, 2012), consistent with atrophy of the hippocampus observed in patients with depression (MacQueen & Frodl, 2011) and often observed in PTSD (Woon, Sood, & Hedges, 2010). SSRIs have both acute and sustained effects on a variety of emotional and cognitive processes (e.g., DiSimplicio, Norbury, & Harmer, 2012; Harmer, 2008). For instance, changes in negative self-schemas in depression have been linked to altered self-referential attention and

positive affect potentiated by SSRI treatment (Harmer, 2008), with comprehensive models highlighting connections between pharmacologically-mediated neurobiological processes and psychological constructs (e.g., Harmer, 2013; Pringle, Browning, Cowen, & Harmer, 2011). Accordingly, although SSRIs do not directly involve provider-targeted efforts to modify traumarelated beliefs, it is quite plausible that these beliefs improve across treatment and may in fact promote PTSD improvement. Indeed, Dozois et al. (2014) found few differences in change on self-reported core beliefs between depressed patients treated with both antidepressants and cognitive therapy versus those treated with antidepressants alone. To our knowledge, no prior study has evaluated the role of trauma-related belief change in symptom improvement in patients with PTSD treated with SSRI monotherapy or compared this relationship to trauma-focused psychotherapy. Thus, our study investigates whether the relationship between belief change and PTSD symptom improvement differs by treatment modality, specifically comparing prolonged exposure versus sertraline, an SSRI with established efficacy for PTSD (Lee et al., 2016).

Cognitive behavioral models of PTSD emphasize the disorder-specific nature of trauma-related beliefs (e.g., Ehlers & Clark, 2000), so changes in these beliefs should mostly clearly impact PTSD symptomatology. Although trauma-related cognitions are most robustly predictive of development of PTSD, they also predict development of major depressive disorder (MDD) to a lesser degree (Kleim, Ehlers, & Glucksman, 2012). MDD co-occurs in more than half of

patients with PTSD (Rytwinski, Scur, Feeny, & Youngstrom, 2013) and depressive symptoms typically improve after treatment with PE (Powers et al., 2010; Ronconi, Shiner & Watts, 2015). The relationship between PTSD and depressive symptoms is complex both in terms of conceptual overlap (Post, Zoellner, Youngstrom, & Feeny, 2011; Zoellner, Pruitt, Farach, & Jun, 2014) and patterns of change across treatment (Aderka, Gillihan, Mclean, & Foa, 2013). Two studies have shown that changes in trauma-related beliefs predict subsequent improvements in depressive symptoms in PE, with results differing in terms of whether this relationship is stronger (Zalta et al., 2014) or weaker (McLean, Yeh et al., 2015) than the link between beliefs and PTSD symptoms. SSRI treatments for PTSD and MDD likely share common pathways, with recent analogue studies suggesting that both fear and mood regulation may be driven by a common neural mechanism (e.g., Tronson et al., 2008). Thus, it is less likely that the relationship between negative-trauma related beliefs and depressive symptom change would differ between PE and sertraline. This comparison also helps to evaluate the specificity of the relationship between trauma-related belief change and PTSD symptom improvement, a key criterion needed to establish a mechanistic effect (Kazdin, 2007).

In this study, we investigated the temporal relationship between change in trauma-related beliefs and symptoms in patients with PTSD treated with PE or sertraline. We utilized time-lagged regression models, which allowed for an estimate of the magnitude of the relationship

between belief and symptom change at the session level, as well as a comparison of the direction of effect (i.e., whether belief change mediates symptom change or vice versa). We hypothesized that belief change would predict subsequent change in PTSD and depressive symptoms in both treatments. We further hypothesized a significant treatment interaction indicating stronger effects in PE than sertraline, in line with the important theoretical role of changing cognitions in PE and some prior evidence of its mediating effects on PTSD symptom improvements. Finally, we expected the relationship between trauma-related belief change and depressive symptom change to be comparable across both PE and sertraline, with no significant moderation by treatment.

Method

Participants

Two hundred participants were recruited for a multi-site trial comparing PE to sertraline in the treatment of PTSD (Zoellner, Roy-Byrne, Mavissakalian, & Feeny, 2017). Institutional review boards at both sites approved the trial. Participants were between 18 and 65 years of age and had a current primary DSM-IV diagnosis of chronic PTSD. Exclusion criteria included: current diagnosis of schizophrenia or other psychotic disorder, unstable bipolar disorder or depression with psychotic features, severe depression requiring immediate treatment (e.g., current actively suicidal with intent); serious self-injurious behavior or suicide attempt within the past three months; no clear trauma memory or primary trauma before age three; alcohol or

substance dependence within the previous three months; ongoing intimate relationship with the perpetrator (in cases of assault); unwilling or medically inadvisable to stop current CBT or pharmacotherapy, based on condition assignment; previous non-response to adequate trial of sertraline (8 weeks, 150mg/d) or PE (8+ sessions); or medical contraindication for initiating sertraline (e.g., pregnant/likely to become pregnant).

For the current study, we developed sample selection criteria based on a relevant theoretical models (Yang & Maxwell. 2014), precedent established by similar studies (e.g., Kleim et al., 2013), and review of patterns of missing data. There is growing recognition that some patients can provide more information about change mechanisms than others (e.g., DeRubeis, Gelfand, German, Fournier, & Forand, 2014), conditional in part on having received some dose of the treatment being investigated. In line with this, most well-powered studies of belief change as a treatment mechanism have focused on patients who attended a specific minimum number of sessions. Thus, patients who are excluded are those contributing little if any data to the primary treatment mechanism questions. This approach also addresses concerns about non-random missingness and estimation accuracy in clinical research models (e.g., Yang & Maxwell. 2014). The present study used the conservative criterion selected by Kleim and colleagues (2013) of including only patients who completed self-report ratings for at least five treatment sessions, resulting in a final sample of 134 patients (n = 83 in PE, n = 51 in sertraline).

Of this sample, 73.9% (n = 99) were missing one or fewer sessions (mean missing = 0.89, median = 0). There was no difference in baseline symptoms or trauma-related cognitions between patients who were excluded and those included in this sample. Demographic and diagnostic characteristics of patients included in these analyses are presented in Table 1. The sertraline condition was slightly older with a mean age of 40.8 (SD = 11.9) versus 36.3 (SD = 11.5) in PE, t(132) = -2.12, p = .04, d = 0.37), with no other differences in demographic characteristics by treatment.

Interview Measures

PTSD Symptom Scale – Interview Version (PSS-I; Foa, Riggs, Dancu & Rothbaum, 1993). The PSS-I is a 17-item clinician rated measure of PTSD symptoms used at baseline to assess PTSD diagnosis and severity for inclusion criteria purposes. This measure has been demonstrated to have good reliability and convergent validity (Foa & Tolin, 2000). In the current study, approximately 10% of cases were re-rated for diagnostic reliability, which was excellent (ICC = .99).

Structural Clinical Interview for the DSM-IV (SCID-IV; First, Spitzer, Gibbon & Williams, 1995). The SCID-IV was used to assess comorbid Axis I diagnoses in this sample for

¹ Missing datapoints represented <8% of the analyzed sample versus >70% for those who were excluded. In the analyzed sample, the modal case of missingness involved a single session in early- or mid-treatment (i.e., not at the end of treatment) likely attributable to failure to fill out questionnaires.

the purposes of inclusion and exclusion criteria.

Self-Report Measures

PTSD Symptom Scale – Self-Report (PSS-SR; Foa et al., 1993). The PSS-SR is a 17item self-report version of the PSS-I. Symptoms are rated on a 4-point scale from 0 (*not at all*) to 3 (*very much*) and are summed to provide a total score. The measure has good reliability and convergent validity (Foa, Cashman, Jaycox & Perry, 1997) and was internally consistent in this sample at pre-treatment, $\alpha = .85$, with a one-week test-retest reliability of .80.

Beck Depression Inventory (BDI; Beck, Ward, Mendelson, Mock & Erbaugh, 1961). The BDI is a 21-item self-report measure of depression, assessing cognitive, affective and vegetative symptoms. Each question includes four response options (0-3), with more severe symptoms reflected in higher scores. The BDI demonstrates good reliability and validity (Beck, Steer & Garbin, 1988) and in the current study at pre-treatment, $\alpha = .88$, with a one-week test-retest reliability of .84.

Posttraumatic Cognitions Inventory (PTCI; Foa et al., 1999). The PTCI is a 36-item self-report measure of negative trauma-related thoughts, with three factors capturing negative beliefs about the self, the world, and self-blame. Each item (e.g., *The world is a dangerous place*) is rated on a scale from 1 (*totally disagree*) to 7 (*totally agree*), with higher scores reflecting more rigid endorsement of negative cognitions. This measure has excellent convergent

and discriminant validity, internal reliability and good test-retest reliability (Foa et al., 1999) and internal consistency at pre-treatment was $\alpha = .95$, with a one-week test-retest reliability of .91

Treatments

PE consisted of 10 weekly sessions, 90-120 min in duration, based on a standard PE treatment manual (Foa, Hembree, & Dancu, 2002). Masters or doctoral-level clinicians who were trained in the delivery of PE provided treatment and received weekly supervision. In sertraline, patients met with board certified psychiatrists for 10 weekly sessions of up to 30 min duration (45 min for the first session). Patients started sertraline at 25 mg/day, with a goal of increasing dose to up to 200 mg/day, as tolerable and clinically indicated using a standard titration algorithm (Marshall, Beebe, Oldham, & Zaninelli, 2001). The average final dose of participants in this subsample was 155 mg/day (SD = 59.0). Treatment adherence ratings were made based on the treatment manual (Foa et al., 2002; Marshall et al., 2001) with 10% of videotapes from the overall study reviewed by trained raters. No protocol violations were detected, and both treatments scored 90% or above on essential treatment components.

Procedure

Participants provided written informed consent prior to enrollment. Participants completed an initial intake evaluation to assess eligibility, at which time they were assessed with the PSS-I and SCID-IV. Eligible participants were assigned to treatment using a computerized

urn randomization stratified by PTSD severity and current anti-depressant status. BDI, PSS-SR, and PTCI ratings were collected at pre-treatment, every treatment session, and post-treatment, and were based on a one-week timeframe, so as to preclude overlapping assessment periods. Although this approach is commonly used in similar treatment research (e.g., Kumpula et al., 2016; Zalta et al., 2014), these measures were originally developed for longer assessment intervals (e.g., 1 month for PTCI; Foa et al., 1999).

Statistical Approach

To test the relationship between symptom change and change in trauma-related beliefs at the session level, we utilized time-lagged, repeated-measures regression (e.g., Rovine & Walls, 2006). This statistical approach allows for an examination of potential temporal relationship between two variables, examining the strength of the relationship between a predictor at Time X, and a dependent variable at Time X+1, while also controlling for the autocorrelation with that predictor at Time X (for additional detail, see Strunk et al., 2012). Scores from the BDI, PSS-SR, and PCTI included up to 12 time points: at pre-treatment, for each of 10 sessions of treatment, and at the post-treatment assessment. For each of these variables, we created a set of dependent variables (Time 2 to Time 12) and a set of lagged predictors (Time 1 to Time 11) combined in a single dataset. All models were tested in PROC Mixed with restricted maximum likelihood in

SAS 9.2 (SAS Institute, Cary NC) to account for missing variables.

We conducted four main sets of analyses: (1) PTCI predicting next-session PSS-SR, (2) PSS-SR predicting next-session PTCI, (3) PTCI predicting next-session BDI, and (4) BDI predicting next-session PTCI. We also included a time covariate, which was significantly predictive in all models, indicating substantial reductions in PSS-SR and PTCI over treatment. Moderator analyses were conducted after first testing for main effects of treatment (PE versus sertraline), which were not significant in any model. In summary, the models predicted time-varying outcome variables by a lagged predictor, covarying time and the outcome autocorrelation (i.e., prior session score).

Results

Belief Change and PTSD Symptom Change

Results of the time-lagged session-to-session models involving PTCI and PSS-SR are presented in Table 2. For both the model of PTSD symptoms and PTCI, the linear effects of time and auto-correlation variables were significant. For the model with belief change as the dependent variable, the cross-lagged effect of PTSD severity (PSS-SR) on negative cognitions (PTCI) was negligible (d < .01), suggesting no relationship between preceding change in PTSD symptoms on subsequent change in beliefs (PTCI). For the model with PTSD symptoms as the dependent variable, the cross-lagged effect of PTCI in predicting subsequent PSS-SR was large

and statistically significant (d = 0.81), suggesting that change in beliefs generally preceded improvements in PTSD symptoms.

Moderation by treatment. In order to examine whether the relationship between belief change and PTSD symptom change differed by treatment modality, we tested Treatment x Predictor interactions in both of the models described above. In the model predicting change in PTCI, the Treatment by PSS-SR interaction was not statistically significant, $\beta = .03$, SE = .08, p = .72. Thus, there were no significant differences between treatments in the relationship between PSS-SR and subsequent PTCI change, which was generally negligible. In the model predicting change in PTSD symptoms, the Treatment x PTCI interaction was significant, $\beta = -.02$, SE = .01, p = .03. Probing this interaction, we examined the relationship between preceding beliefs change and next-session PTSD symptoms in each treatment separately. In sertraline, PTCI was associated with subsequent PTSD symptom improvement at a trend level with a small effect size, $\beta = .02$, SE = .01, p = .08, d = 0.35. In PE, the relationship was considerably stronger, with a large effect size ($\beta = .05$, SE = .01, p < .0001, d = 0.93). Therefore, the interaction reflects a difference in magnitude between treatments, with a small trend-level relationship of belief change preceding PTSD symptom change in sertraline versus a much more robust relationship in PE of belief change preceding PTSD symptom change.

Belief Change and Depressive Symptom Change

Table 3 shows results for the time-lagged session-to-session models involving traumarelated negative beliefs (PTCI) and depression (BDI) change. As in the PTSD symptom models, the linear effects of time and auto-correlation variables were significant in both models. For the model predicting change in PTCI, the cross-lagged effect of BDI on PTCI was statistically significant with a small to medium effect size (d = 0.30), indicating that preceding depressive symptoms were modestly related to subsequent belief change. For the model with BDI as the dependent variable, cross-lagged PTCI scores were significantly predictive of subsequent BDI scores with a medium effect size (d = 0.49), suggesting that change in beliefs also preceded improvements in depression symptoms.

Moderation by treatment. We tested Treatment x Predictor interactions in both of the belief change and depressive symptom models. In the model predicting change in PTCI, the Treatment x BDI interaction was not a significant predictor of belief change (β = .05, SE = .08, p = .54). In the model predicting change in BDI, the Treatment x PTCI interaction was not significant in predicting subsequent BDI change (β = .01, SE = .01, p = .42), indicating that the relationship between depressive symptom improvement and belief change did not differ by treatment modality. Thus, unlike our PTSD symptom findings, there were no significant differences between treatments in the relationship between BDI and PTCI change, regardless of the direction of lagged effect being modeled.

Discussion

Trauma-related belief change preceded PTSD symptom improvement more robustly in patients treated with PE versus those treated with sertraline. As hypothesized, the relationship between belief change and depression improvement was less substantial, with no clear direction of effect, and no difference between treatments. The robust link between belief change and PTSD change in PE is consistent with the emphasis on negative trauma-related beliefs in CBT treatments and related psychological models (Ehlers & Clark, 2000; Foa et al., 2006). Patients who received sertraline also saw reductions in trauma-related beliefs across treatment, temporally preceding PTSD symptom improvement, but this relationship was small to moderate.

These findings are consistent with emotional processing theory in highlighting belief change as a precursor to PTSD recovery (Foa et al., 2006), regardless of how it occurs. Although belief change occurs with SSRIs, this treatment likely promotes symptom reduction via more general neurochemical changes in fear circuitry, consistent with the broad changes in functioning in response to SSRIs (Brady et al., 2000; Davidson et al., 2001). This is similar to the framework described by Mayberg and colleagues (Mayberg et al., 1999), whereby psychotherapies exert "top down" effects on symptomatology (i.e., cognitive / cortical processes regulating limbic response) versus more "bottom-up" or mixed effects of SSRIs (i.e., affecting brainstem nuclei with secondary effects on cortical sites). Models of SSRI treatment for mood and anxiety

disorders have not traditionally emphasized a role for change in specific psychological processes, and only recently have efforts been made to incorporate these factors into cognitive neuropsychological models of treatment effects (e.g., Harmer, 2013; Pringle et al., 2011) or test modality-specific differences in their relation to treatment change (e.g., Dozois et al., 2014). SSRI-mediated neurological changes in emotionally-biased attention and self-referential processing (e.g., DiSimplicio et al., 2012) may promote more approach-oriented behaviors, which in turn could provide disconfirming evidence with respect to strongly held, erroneous negative beliefs. Alternatively, self-reported belief change may be an epiphenomenon of other change processes (e.g., Hofmann, 2008), such as reversal of neuronal deficits in key brain areas (Duman & Voleti, 2012), constituting a different level of measurement with respect to the neurobiological mechanisms by which SSRI treatment achieves its effects (Harmer, 2013)². Regardless, most notably, these findings suggest that the change in negative beliefs is a more relevant marker of PTSD improvement in PE versus SSRI treatment.

The parallel examination of depressive symptoms provides an important comparison to the primary models of PTSD change. Notably, associations between trauma-related belief change and depression were more modest than those between PTSD and beliefs, consistent with the

² We take care not to describe biological mechanisms as *underlying* belief change, or vice versa. See Miller (2010) for a detailed discussion of the potential problems with this term.

trauma-specific ways of thinking emphasized by cognitive models of PTSD (e.g., Ehlers & Clark, 2000). Although certain symptoms of PTSD and MDD overlap (e.g., insomnia; Post et al., 2011), prior research has typically found stronger predictive relationships between traumarelated negative beliefs and PTSD symptoms versus depressive symptoms (Kleim et al., 2012). These findings also suggest change in beliefs and depressive symptoms had a more reciprocal temporal relationship, with belief change predicting subsequent depressive symptom improvement and vice versa. Most critically, the relationship between beliefs and depressive symptoms did not significantly differ between PE and sertraline, regardless of which temporal sequence was being tested. Thus, whereas the relationship between trauma-related belief change and PTSD symptom change was clearly unidirectional and considerably stronger in PE versus sertraline, the relationship between belief change and depressive symptom analyses were more modest, reciprocal, and comparable across treatments. This is in line with the specific role ascribed to negative trauma-related beliefs in the maintenance of PTSD symptoms in several cognitive behavioral models of PTSD (e.g., Foa et al., 2006; Hoffmann, 2008). These findings suggest trauma-related belief change likely exerts a more robust influence on PTSD symptoms, and perhaps especially so in PE.

Key strengths of this study include the theoretically important and unique comparison between PE and sertraline, the conservative analytic strategy, and use of a large and diverse sample of male and female patients with heterogeneous target traumas. However, we strongly caution readers not to draw strong, causal inferences about the relationships depicted in these models, as these are fundamentally correlational. Although we utilize a conservative statistical method that controls for prior change and establishes temporal precedence, it is possible that common factors (e.g., supportive contact with study personnel) or modality-specific factors (e.g., content of in-session processing in PE) may help to explain both change in trauma-related cognitions and symptom change. Similarly, these analyses do not inform us as to how belief change may be occurring – that is, what precipitates or motivates such change. Other limitations of this study include unequal size of PE and sertraline conditions, a consequence of the design of the clinical trial from which data are drawn. This study also used DSM-IV symptom criteria for PTSD. As noted by Schumm and colleagues (2013), in the DSM-5 (American Psychiatric Association, 2013) cognitive alterations are now considered a symptom of PTSD. Future studies using DSM-5 criteria will need to carefully consider whether trauma-related negative beliefs are better represented as cognitive mechanisms, sequelae, or symptoms of PTSD. It is possible that future work in this area will be framed in terms of sequential symptom change rather than mechanisms per se.

Our analyses focus on total scores of symptom and belief measures (i.e., PTCI, PSS-SR, BDI), as these had the most robust prior support and were most likely to be relevant to tests of

moderation by treatment. As such, our findings do not speak to potential differences by subscales or symptom clusters, as highlighted by a recent study showing stronger effects for self-relevant trauma beliefs on PTSD gains (Kumpula et al., 2016). However, testing all possible combinations of PTCI subscales and PSS-SR clusters across our four main models would greatly raise the risk of a Type I error. Self-report measures were also adjusted to reflect a one-week timespan interval, departing from longer durations specified during development of these measures (e.g., Foa et al., 1993; Foa et al., 1999). Although this shorter interval is commonly used in clinical research on belief change and PTSD (e.g., Zalta et al., 2014; Kumpula et al., 2016), its influence on the psychometrics of these scales and measurement of processes across time is not well established. However, the relationship between PTCI and PTSD symptoms has been quite consistent across studies of PE despite method variance in assessments, including studies separating assessment of beliefs and symptoms (e.g., Oktedalen et al., 2014) or using longer intervals between assessments (e.g., 4 weeks; McLean et al., 2015). Finally, as most prior studies of this kind have done (e.g., Kleim et al., 2013), we chose to focus on patients who attended a minimum number of sessions, based on contemporary guidance for mechanism research (Cohen & DeRubeis, 2016; DeRubeis et al., 2014). The analyzed sample had very little missing data, whereas patients excluded from analyses often had more extensive missing data, following patterns characteristic of attrition (i.e., consecutive missing sessions at the end of

treatment). Accordingly, the present findings are most relevant to understanding change processes in patients who engaged with treatment and do not represent the entire randomized sample.

In summary, trauma-related belief change is linked to PTSD improvement across modalities yet more robustly predictive in cognitive behavioral therapy than SSRI treatment with sertraline, suggestive of different processes implicated PTSD symptom improvement. Notably, the relationship between belief change and depressive symptoms was not moderated by treatment nor clearly directional, highlighting the relatively more robust link between trauma-related cognitive change and PTSD improvements. Psychotherapy process research may illuminate the elements of treatment most strongly associated with belief change and suggest ways to enhance this process during treatment, such as by tailoring between-session activities (i.e., in vivo homework assignments) to focus on persistent negative beliefs. Similarly, novel research paradigms such as experience-sampling might offer insight into patients' subjective experience of changes in negative beliefs and PTSD improvement across shorter time intervals. Beyond their significance to theory and research on change mechanisms of PE and SSRIs, our findings also have clear clinical implications. In PE, failure to experience trauma-related belief change may be a warning sign of an impending clinical impasse or stagnation. Furthermore, belief change may be a critical pathway to overcoming such setbacks and enhancing treatment

outcomes. Beyond routinely assessing beliefs across treatment, further study is needed as to the best approach to address such situations clinically within the context of PE. Strategies could include explicitly tracking change in each patients' most troubling or central negative belief as suggested by Kleim et al. (2013) or incorporating more explicit cognitive change techniques (i.e., cognitive restructuring; CR) from other trauma-focused CBTs during post-imaginal processing. However, the addition of CR to PE protocols does not enhance overall treatment response (Jonas et al., 2013). In fact, one study found that patients with more severe negative beliefs at baseline showed a worse response to combined PE plus CR than conventional PE (Moser, Cahill, & Foa, 2010). Although the addition of cognitive techniques may not be warranted for all cases, an additional possibility is that certain patients would benefit from these approaches, though who these patients are is not clear. Research on moderation of treatment mechanisms is lacking at present in the PTSD literature, but there is evidence from the depression literature that certain patient traits predict more robust response to cognitive techniques, including gender (e.g., Sasso, Strunk, Braun, DeRubeis & Brotman, 2015). Further study is needed in this area, and in terms of potential cascading temporal relationships between belief change, PTSD symptoms and depressive symptoms across treatment. In summary, this work adds to the growing body of research highlighting trauma-related beliefs as predictors of change in trauma-focused CBTs (Kleim et al., 2013; Kumpula et al., 2015; McLean, Su et al., 2015; Oktedalen et al., 2014;

Cooper, Zoellner et al. (2017)

Schumm et al., 2013; Zalta et al., 2014) and further suggests that different change processes may promote improvement in sertraline treatment.

References

- Aderka, I. M., Gillihan, S. J., McLean, C. P., & Foa, E. B. (2013). The relationship between posttraumatic and depressive symptoms during prolonged exposure with and without cognitive restructuring for the treatment of posttraumatic stress disorder. *Journal of Consulting and Clinical Psychology*, 81(3), 375-382. doi:10.1037/a0031523
- American Psychiatric Association. (2013). *Diagnostic and statistical manual of mental disorders* (5th ed.). Arlington, VA: American Psychiatric Publishing.
- Beck, A. T., Emery, G., & Greenberg, R. L. (2005). *Anxiety disorders and phobias: A cognitive perspective*. New York: Basic Books.
- Beck, A. T., Steer, R. A., & Garbin, M. G. (1988). Psychometric properties of the Beck

 Depression Inventory: Twenty-five years of evaluation. *Clinical Psychology*Review, 8(1), 77-100. doi:10.1016/0272-7358(88)90050-5
- Beck, A.T., Ward, C. H., Mendelson, M., Mock, J., & Erbaugh, J. (1961). An inventory for measuring depression. *Archives of General Psychiatry*, 4, 561-571. doi:10.1001/archpsyc.1961.01710120031004
- Bouton, M. (1991). A contextual analysis of fear extinction. In P. R. Martin, (Ed.), *Handbook*

- of behavior therapy and psychological science: An integrative approach (pp. 435-453). Elmsford, NY: Pergamon Press.
- Bouton, M. (2004). Context and behavioral process in extinction. *Learning & Memory*, 11, 485-494. doi:10.1101/lm.78804
- Bruce, S. E., Yonkers, K. A., Otto, M. W., Eisen, J. L., Weisberg, R. B., Pagano, M., ... & Keller, M. B. (2008). Influence of psychiatric comorbidity on recovery and recurrence in generalized anxiety disorder, social phobia, and panic disorder: a 12-year prospective study. *American Journal of Psychiatry*, *162*, 1179–1187. doi:10.1176/appi.ajp.162.6.1179.
- Bryant, R. A., & Guthrie, R. M. (2007). Maladaptive self-appraisals before trauma exposure predict posttraumatic stress disorder. *Journal of Consulting and Clinical Psychology*, 75, 812-815. doi:10.1037/0022-006X.75.5.812
- Cohen, J. A., & Mannarino, A. P. (1996). *Client-centered therapy treatment manual*. Pittsburgh, PA: MCP Hahnemann University School of Medicine (Unpublished treatment manual).
- Cohen, Z.D. & DeRubeis, R.D. (2016). Selecting the right patients for understanding moderators and process-outcome relationships in order to guide treatment selection. In N. Garcia (Chair), *Innovative Predictors of Treatment Outcome*. Presented at the 50th annual meeting of the Association for Behavioral and Cognitive Therapies. New York, NY.

- Craske, M. G., Treanor, M., Conway, C. C., Zbozinek, T., & Vervliet, B. (2014). Maximizing exposure therapy: An inhibitory learning approach. *Behaviour Research and Therapy*, 58, 10-23. doi:10.1016/j.brat.2014.04.006
- Cuthbert, B. N., & Insel, T. R. (2013). Toward the future of psychiatric diagnosis: The seven pillars of RDoC. *BMC Medicine*, *11*(1), 126. doi:10.1186/1741-7015-11-126
- Davidson, J. R., Landerman, L. R., Farfel, G. M., & Clary, C. M. (2002). Characterizing the effects of sertraline in post-traumatic stress disorder. *Psychological Medicine*, *32*(04), 661-670. doi:10.1017/S0033291702005469
- DeRubeis, R. J., Gelfand, L. A., German, R. E., Fournier, J. C., & Forand, N. R. (2014).

 Understanding processes of change: How some patients reveal more than others—and some groups of therapists less—about what matters in psychotherapy. *Psychotherapy Research*, 24(3), 419-428. doi: 10.1080/10503307.2013.838654
- DiSimplicio, M., Norbury, R., & Harmer, C. J. (2012). Short-term antidepressant administration reduces negative self-referential processing in the medial prefrontal cortex in subjects at risk for depression. *Molecular Psychiatry*, *17*(5), 503-510. doi:10.1038/mp.2011.16
- Dozois, D. J., Bieling, P. J., Evraire, L. E., Patelis-Siotis, I., Hoar, L., Chudzik, S., ... & Westra, H. A. (2014). Changes in core beliefs (early maladaptive schemas) and self-representation in cognitive therapy and pharmacotherapy for depression. *International*

- Journal of Cognitive Therapy, 7(3), 217-234. doi: 10.1521/ijct.2014.7.3.217
- Duman, R. S., & Voleti, B. (2012). Signaling pathways underlying the pathophysiology and treatment of depression: novel mechanisms for rapid-acting agents. *Trends in Neurosciences*, 35(1), 47-56. doi:10.1016/j.tins.2011.11.004
- Dunmore, E., Clark, D. M., & Ehlers, A. (1999). Cognitive factors involved in the onset and maintenance of posttraumatic stress disorder (PTSD) after physical or sexual assault. *Behaviour Research and Therapy*, *37*(9), 809-829. doi:10.1016/S0005-7967(98)00181-8
- Ehlers, A., & Clark, D. M. (2000). A cognitive model of posttraumatic stress disorder. *Behaviour Research and Therapy*, 38(4), 319-345. doi: 10.1016/S0005-7967(99)00123-0
- Ehring, T., Ehlers, A., & Glucksman, E. (2008). Do cognitive models help in predicting the severity of posttraumatic stress disorder, phobia, and depression after motor vehicle accidents? A prospective longitudinal study. *Journal of Consulting and Clinical Psychology*, 2, 219-230. doi:10.1017/S0033291707001006
- First, M. B., Spitzer, R. L., Williams, J. B. W., & Gibbon, M. (1995). Structured Clinical Interview for DSM-IV (SCID). Washington, DC: American Psychiatric Association.
- Foa, E. B., Cashman, L., Jaycox, L., & Perry, K. (1997). The validation of a self-report measure of posttraumatic stress disorder: the Posttraumatic Diagnostic Scale. *Psychological*

- Assessment, 9(4), 445-451. doi:10.1037/1040-3590.9.4.445
- Foa, E. B., Ehlers, A., Clark, D. M., Tolin, D. G., & Orsillo, S. M. (1999). The Posttraumatic Cognitions Inventory (PTCI): Development and validation. *Psychological Assessment*, 11, 303–314. doi:10.1037/1040-3590.11.3.303
- Foa, E. B., Hembree, E. A., Cahill, S. P., Rauch, S. A., Riggs, D. S., Feeny, N. C., & Yadin, E. (2005). Randomized trial of prolonged exposure for PTSD with and without cognitive restructuring: Outcome at academic and community clinics. *Journal of Consulting and Clinical Psychology*, 73, 953-964. doi: 10.1037/0022-006X.73.5.953
- Foa, E.B., Hembree, E.A., & Dancu, C.V. (2002). *Prolonged Exposure (PE) Manual: Revised Version*. Unpublished manuscript.
- Foa, E.B., Huppert, J.D., & Cahill, S.P. (2006). Emotional processing theory: An update. In Rothbaum, B.O. (Ed.), *Pathological Anxiety: Emotional processing in etiology and treatment.* (pp. 3-24). New York: Guilford Press.
- Foa, E. B., & Kozak, M. J. (1986). Emotional processing of fear: Exposure to corrective information. *Psychological Bulletin*, *99*, 20–35. doi: 10.1037/0033-2909.99.1.20
- Foa, E. B., & McLean, C. P. (2016). The efficacy of exposure therapy for anxiety-related disorders and its underlying mechanisms: The case of OCD and PTSD. *Annual Review of Clinical Psychology*, *12*, 1-28. doi: 10.1146/annurev-clinpsy-021815-093533

- Foa, E. B., & Rauch, S. A. (2004). Cognitive changes during prolonged exposure versus prolonged exposure plus cognitive restructuring in female assault survivors with posttraumatic stress disorder. *Journal of Consulting and Clinical Psychology*, 72(5), 879-884. doi: 10.1037/0022-006X.72.5.879
- Foa, E., Riggs, D., Dancu, C., & Rothbaum, B.O. (1993). Reliability and validity of a brief instrument for assessing post-traumatic stress disorder. *Journal of Traumatic Stress*, 6, 459-473. doi:10.1002/jts.2490060405
- Foa, E. B., & Rothbaum, B. O. (1998). *Treating the trauma of rape: Cognitive behavioral therapy for PTSD*. New York: Guilford Press.
- Foa, E. B., & Tolin, D. F. (2000). Comparison of the PTSD Symptom Scale–Interview version and the Clinician-Administered PTSD scale. *Journal of Traumatic Stress*, *13*(2), 181-191. doi: 10.1023/A:1007781909213
- Hagenaars, M. A., Van Minnen, A., & de Rooij, M. (2010). Cognitions in prolonged exposure therapy for posttraumatic stress disorder. *International Journal of Clinical and Health Psychology*, 10(3), 421-434. doi: 10.1037/a0031290
- Harmer, C. J. (2008). Serotonin and emotional processing: does it help explain antidepressant drug action?. *Neuropharmacology*, 55(6), 1023-1028. doi:10.1016/j.neuropharm.2008.06.036

- Harmer, C. J. (2013). Emotional processing and antidepressant action. *Current Topics in Behavioral Neuroscience*, *14*, 209-222. DOI: 10.1007/7854_2012_210
- Hirschfeld, R. M. A., Montgomery, S. A., Aguglia, E., Amore, M., Delgado, P. L., Gastpar, M., ... Versianix, M. (2002). Partial response and nonresponse to antidepressant therapy. *Journal of Clinical Psychiatry*, 63, 826–837. doi:10.4088/jcp.v63n0913
- Hofmann, S. G. (2008). Cognitive processes during fear acquisition and extinction in animals and humans: Implications for exposure therapy of anxiety disorders. *Clinical Psychology Review*, 28(2), 199-210. doi:10.1016/j.cpr.2007.04.009
- Hofmann, S. G., Ellard, K. K., & Siegle, G. J. (2012). Neurobiological correlates of cognitions in fear and anxiety: a cognitive–neurobiological information-processing model. *Cognition & Emotion*, 26(2), 282-299. doi: 10.1080/02699931.2011.579414
- Hofmann, S. G., Asnaani, A., Vonk, I. J., Sawyer, A. T., & Fang, A. (2012). The efficacy of cognitive behavioral therapy: A review of meta-analyses. *Cognitive Therapy and Research*, *36*, 427-440. doi:10.1007/s10608-012-9476-1
- Hollon, S. D., Jarrett, R. B., Nierenberg, A. A., Thase, M. E., Trivedi, M., & Rush, A. J. (2005).
 Psychotherapy and medication in the treatment of adult and geriatric depression: which monotherapy or combined treatment?. *Journal of Clinical Psychiatry*, 66(4), 455-468.
 http://dx.doi.org/10.4088/jcp.v66n0408

- Jonas, D. E., Cusack, K., Forneris, C. A., Wilkins, T. M., Sonis, J., Middleton, J. C., Gaynes, B. N. (2013). *Psychological and pharmacological treatments for adults with posttraumatic stress disorder (PTSD)* (Report No: 13-EHC011-EF). Rockville, MD: Agency for Healthcare Research and Quality.
- Kazdin, A. E. (2007). Mediators and mechanisms of change in psychotherapy research. *Annual Review of Clinical Psychology*, *3*, 1-27. doi:10.1146/annurev.clinpsy.3.022806.091432
- Kleim, B., Ehlers, A., & Glucksman, E. (2012). Investigating cognitive pathways to psychopathology: Predicting depression and posttraumatic stress disorder from early responses after assault. *Psychological Trauma: Theory, Research, Practice, and Policy*, 4, 527 537. doi: 10.1037/a0027006
- Kleim, B., Grey, N., Wild, J., Nussbeck, F. W., Stott, R., Hackmann, A., Clark, D. M., & Ehlers, A. (2013). Cognitive change predicts symptom reduction with cognitive therapy for posttraumatic stress disorder. *Journal of Consulting and Clinical Psychology*, 81(3), 383-393. doi:10.1037/a0031290
- Kumpula, M. J., Pentel, K. Z., Foa, E. B., LeBlanc, N. J., Bui, E., McSweeney, L. B., ... & Rauch, S. A. (2016). Temporal sequencing of change in posttraumatic cognitions and PTSD symptom reduction during prolonged exposure therapy. *Behavior Therapy*. Advanced online publication. doi:10.1016/j.beth.2016.02.008

- Lancaster, S. L., Rodriguez, B. F., & Weston, R. (2011). Path analytic examination of a cognitive model of PTSD. *Behaviour Research and Therapy*, 49, 194-201. doi:10.1016/j.brat.2011.01.002
- Lang, P. J. (1979). A bio- informational theory of emotional imagery. *Psychophysiology*, *16*(6), 495-512. doi: 10.1111/j.1469-8986.1979.tb01511.x
- Lee, D. J., Schnitzlein, C. W., Wolf, J. P., Vythilingam, M., Rasmusson, A. M., & Hoge, C. W. (2016). Psychotherapy versus pharmacotherapy for posttraumatic stress disorder:

 Systemic review and meta- analyses to determine first- line treatments. *Depression & Anxiety*. 33, 792-806. doi: 10.1002/da.22511
- LoSavio, S. T., Dillon, K. H., & Resick, P. A. (2017). Cognitive factors in the development, maintenance, and treatment of post-traumatic stress disorder. *Current Opinion in Psychology*, *14*, 18-22. doi: 10.1016/j.copsyc.2016.09.006
- MacQueen, G., & Frodl, T. (2011). The hippocampus in major depression: Evidence for the convergence of the bench and bedside in psychiatric research? *Molecular Psychiatry*, 16(3), 252-264. doi:10.1038/mp.2010.80
- Marshall, R. D., Beebe, K. L., Oldham, M., & Zaninelli, R. (2001). Efficacy and safety of paroxetine treatment for chronic PTSD: A fixed-dose, placebo-controlled study. *The American Journal of Psychiatry*, 158(12), 1982-1988. doi:10.1176/appi.ajp.158.12.1982

- Mayberg, H. S., Liotti, M., Brannan, S. K., McGinnis, S., Mahurin, R. K., Jerabek, P. A., ... & Fox, P. T. (1999). Reciprocal limbic-cortical function and negative mood: Converging PET findings in depression and normal sadness. *American Journal of Psychiatry*, 156, 675-682. doi: 10.1176/ajp.156.5.675
- McLean, C. P., Su, Y. J., & Foa, E. B. (2015). Mechanisms of symptom reduction in a combined treatment for comorbid posttraumatic stress disorder and alcohol dependence. *Journal of Consulting and Clinical Psychology*, 83(3), 655-661. doi::10.1037/ccp0000024
- McLean, C. P., Yeh, R., Rosenfield, D., & Foa, E. B. (2015). Changes in negative cognitions mediate PTSD symptom reductions during client-centered therapy and prolonged exposure for adolescents. *Behaviour Research and Therapy*, 68, 64-69. doi:10.1016/j.brat.2015.03.008
- McNally, R. J. (2007). Mechanisms of exposure therapy: How neuroscience can improve psychological treatments for anxiety disorders. *Clinical Psychology Review*, 27(6), 750-759. doi:10.1016/j.cpr.2007.01.003
- Miller, G.A. (2010). Mistreating psychology in the decades of the brain. *Perspectives on Psychological Science*, 5,716-743. doi: 10.1177/1745691610388774
- Moser, J. S., Cahill, S. P., & Foa, E. B. (2010). Evidence for poorer outcome in patients with severe negative trauma-related cognitions receiving prolonged exposure plus cognitive

- restructuring: Implications for treatment matching in posttraumatic stress disorder. *The Journal of Nervous and Mental Disease*, 198(1), 72-75. doi: 10.1097/NMD.0b013e3181c81fac
- Mueller, T. I., Leon, A. C., Keller, M. B., Solomon, D. A., Endicott, J., Coryell, W. ... & Maser, J.D. (1999). Recurrence after recovery from major depressive disorder during 15 years of observational follow-up. *American Journal of Psychiatry*, *156*, 1000–1006.
- Mueser, K. T., Rosenberg, S. D., Xie, H., Jankowski, M. K., Bolton, E. E., Lu, W., ... & Wolfe,
 R. (2008). A randomized controlled trial of cognitive-behavioral treatment for
 posttraumatic stress disorder in severe mental illness. *Journal of Consulting and Clinical Psychology*, 76(2), 259-271. doi: 10.1037/0022-006X.76.2.259
- Murphy, R., Cooper, Z., Hollon, S. D., & Fairburn, C. G. (2009). How do psychological treatments work? Investigating mediators of change. *Behaviour Research and Therapy*, 47(1), 1-5. doi:10.1016/j.brat.2008.10.001
- Nacasch, N., Huppert, J. D., Su, Y. J., Kivity, Y., Dinshtein, Y., Yeh, R., & Foa, E. B. (2015).

 Are 60-minute Prolonged Exposure sessions with 20-minute imaginal exposure to traumatic memories sufficient to successfully treat PTSD? A randomized noninferiority clinical trial. *Behavior Therapy*, 46, 328-341. doi:10.1016/j.beth.2014.12.002
- Nathan, P. E., & Gorman, J. M. (2015). A guide to treatments that work, 4th edition. Newy York,

- New York: Oxford University Press.
- Øktedalen, T., Hoffart, A., & Langkaas, T. F. (2015). Trauma-related shame and guilt as time-varying predictors of posttraumatic stress disorder symptoms during imagery exposure and imagery rescripting—A randomized controlled trial. *Psychotherapy Research*, 25, 1-15. doi: 10.1080/10503307.2014.917217
- Post, L. M., Zoellner, L. A., Youngstrom, E., & Feeny, N. C. (2011). Understanding the relationship between co-occurring PTSD and MDD: Symptom severity and affect. *Journal of Anxiety Disorders*, 25(8), 1123-1130. doi:10.1016/j.janxdis.2011.08.003
- Powers, M. B., Halpern, J. M., Ferenschak, M. P., Gillihan, S. J., & Foa, E. B. (2010). A metaanalytic review of prolonged exposure for posttraumatic stress disorder. *Clinical Psychology Review*, *30*(6), 635-641. doi: 10.1016/j.cpr.2010.04.007
- Pringle, A., Browning, M., Cowen, P. J., & Harmer, C. J. (2011). A cognitive neuropsychological model of antidepressant drug action. *Progress in Neuro-Psychopharmacology and Biological Psychiatry*, *35*(7), 1586-1592. doi:10.1016/j.pnpbp.2010.07.022
- Resick, P. A., Nishith, P., Weaver, T. L., Astin, M. C., & Feuer, C. A. (2002). A comparison of cognitive-processing therapy with prolonged exposure and a waiting condition for the

- treatment of chronic posttraumatic stress disorder in female rape victims. *Journal of Consulting and Clinical Psychology*, 70(4), 867-879. doi: 10.1037/0022-006X.70.4.867
- Ronconi, J. M., Shiner, B., & Watts, B. V. (2015). A meta-analysis of depressive symptom outcomes in randomized, controlled trials for PTSD. *The Journal of Nervous and Mental Disease*, 203(7), 522-529. doi: 10.1097/NMD.0000000000000022
- Rovine, M. J., & Walls, T. A. (2006) Multilevel autoregressive modeling of interindividual differences in the stability of a process. In T. A. Walls & J. L. Schafer (Eds.), *Models for Intensive Longitudinal Data* (pp. 124-147). New York: Oxford University Press.
- Rytwinski, N. K., Scur, M. D., Feeny, N. C., & Youngstrom, E. A. (2013). The co-occurrence of major depressive disorder among individuals with posttraumatic stress disorder: A meta-analysis. *Journal of Traumatic Stress*, 26(3), 299-309. doi:10.1002/jts.21814
- Sasso, K. E., Strunk, D. R., Braun, J. D., DeRubeis, R. J., & Brotman, M. A. (2015). Identifying moderators of the adherence-outcome relation in cognitive therapy for depression. *Journal of Consulting and Clinical Psychology*, 83, 976-984. doi: 10.1037/ccp0000045
- Schumm, J. A., Dickstein, B. D., Walter, K. H., Owens, G. P., & Chard, K. M. (2015).

 Changes in posttraumatic cognitions predict changes in posttraumatic stress disorder symptoms during Cognitive Processing Therapy. *Journal of Consulting and Clinical*

- Psychology. 83, 1161-1166. doi:10.1037/ccp0000040
- Shahar, G., Noyman, G., Schnidel-Allon, I., & Gilboa-Schechtman, E. (2013). Do PTSD symptoms and trauma-related cognitions about the self constitute a vicious cycle? Evidence for both cognitive vulnerability and scarring models. *Psychiatry Research*, 205(1), 79-84. doi:10.1016/j.psychres.2012.07.053
- Strunk, D. R., Cooper, A. A., Ryan, E. T., DeRubeis, R. J., & Hollon, S. D. (2012). The process of change in cognitive therapy for depression when combined with antidepressant medication: Predictors of early intersession symptom gains. *Journal of Consulting and Clinical Psychology*, 80(5), 730-738. doi: 10.1037/a0029281
- Taylor, S., Abramowitz, J. S., & McKay, D. (2012). Non-adherence and non-response in the treatment of anxiety disorders. *Journal of Anxiety Disorders*, 26, 583-589. doi:10.1016/j.janxdis.2012.02.010
- Taylor, S., Thordarson, D. S., Maxfield, L., Fedoroff, I. C., Lovell, K., & Ogrodniczuk, J. (2003). Comparative efficacy, speed, and adverse effects of three PTSD treatments: exposure therapy, EMDR, and relaxation training. *Journal of Consulting and Clinical Psychology*, 71(2), 330-338. doi: 10.1037/0022-006X.71.2.330
- Tronson, N. C., Schrick, C., Fischer, A., Sananbenesi, F., Pagès, G., Pouysségur, J., & Radulovic, J. (2008). Regulatory mechanisms of fear extinction and depression-like

- behavior. *Neuropsychopharmacology*, 33(7), 1570-1583. doi:10.1038/sj.npp.1301550
- Woon, F. L., Sood, S., & Hedges, D. W. (2010). Hippocampal volume deficits associated with exposure to psychological trauma and posttraumatic stress disorder in adults: a meta-analysis. *Progress in Neuro-Psychopharmacology and Biological Psychiatry*, *34*(7), 1181-1188. doi:10.1016/j.pnpbp.2010.06.016
- Yang, M., & Maxwell, S. E. (2014). Treatment effects in randomized longitudinal trials with different types of nonignorable dropout. *Psychological Methods*, *19*(2), 188-210. doi: 10.1037/a0033804
- Zalta, A. K., Gillihan, S. J., Fisher, A. J., Mintz, J., McLean, C. P., Yehuda, R., & Foa, E. B.
 (2014). Change in negative cognitions associated with PTSD predicts symptom reduction in prolonged exposure. *Journal of Consulting and Clinical Psychology*, 82(1), 171-175.
 doi: 10.1037/a0034735
- Zoellner, L. A., Feeny, N. C., Eftekhari, A., & Foa, E. B. (2011). Changes in negative beliefs following three brief programs for facilitating recovery after assault. *Depression and Anxiety*, 28(7), 532-540. doi: 10.1002/da.20847
- Zoellner, L.A., Roy-Byrne, P.B., Mavissakalian, M,. & Feeny, N.C. *Doubly randomized*preference trial in PTSD: Prolonged Exposure versus sertraline. Manuscript submitted for publication.

Zoellner, L. A., Pruitt, L. D., Farach, F. J., & Jun, J. J. (2014). Understanding heterogeneity in PTSD: Fear, dysphoria, and distress. *Depression and Anxiety*, *31*(2), 97-106. doi: 10.1002/da.22133